**Author’s response to reviews**

**Title:** Association of physical activity with lung function in lung-healthy German adults: Results from the KORA FF4 study

**Authors:**

Agnes Luzak (agnes.luzak@helmholtz-muenchen.de)

Stefan Karrasch (stefan.karrasch@helmholtz-muenchen.de)

Barbara Thorand (thorand@helmholtz-muenchen.de)

Dennis Nowak (dennis.nowak@med.uni-muenchen.de)

Rolf Holle (holle@helmholtz-muenchen.de)

Annette Peters (peters@helmholtz-muenchen.de)

Holger Schulz (schulz@helmholtz-muenchen.de)

**Version:** 2  **Date:** 14 Dec 2017

**Author’s response to reviews:**

Dear Editor,

Thank you for giving us the opportunity to re-submit our manuscript entitled “Association of physical activity with lung function in lung-healthy German adults: Results from the KORA FF4 study” for consideration in BMC Pulmonary Medicine. We are very grateful for your and the reviewers additional comments and scientific information. We applied all formal changes and hope that the updated version meets the criteria for publication in your Journal.

Thank you for your consideration.

Sincerely,

Holger Schulz